Author's response to reviews

Title: Motivational Interviewing as an Intervention to Increase Adolescent Self-Efficacy and Promote Weight loss

Authors:

Beverly Walpole (bwalpole@uoguelph.ca)
Elizabeth Dettmer (elizabeth.dettmer@sickkids.ca)
Barbara Morrongiello (bmorrong@uoguelph.ca)
Brian McCrindle (brian.mccrindle@sickkids.ca)
Jill Hamilton (jill.hamilton@sickkids.ca)

Version: 4 Date: 3 May 2011

Author's response to reviews:

Revisions made on May 2, 2011:

1. Shortened the length of the abstract within the submission system to be 340 words and identical to the abstract within the body of the manuscript.